Author’s response to reviews

Title: Posterior decompression and occipitocervical fixation followed by intraoperative vertebroplasty for metastatic involvement of the axis

Authors:

Xinjie WU (wuxinjie_ms@163.com)
Yingna QI (ayingqi119925@163.com)
mingsheng tan (doctortan_123@126.com)
Ping YI (dryiping_123@163.com)
Feng YANG (dryangfeng@126.com)
Xiangsheng TANG (drtangxiangsheng@126.com)
Qingying HAO (drhaoqingying@126.com)

Version: 4 Date: 28 Dec 2017

Author’s response to reviews:

1. Please include the email addresses for all authors on the title page. We also have noticed that the email address for the corresponding author in the manuscript differs from the email address provided in the submission system. Please also ensure all email addresses provided in the manuscript and the submission system match.

Authors: Email addresses for all authors has been provided in P1L11-17.

2. In the Ethics approval and consent to participate section of the declarations, please include the FULL name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

Please also confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Authors: The full name of the ethics committee has been provided in P1L233-234. Written informed consent was obtained from all participants and this has been added in P1L236.
3. We note that you have included an image of the patient and age and gender details in the figure legend. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. Please state whether you have obtained consent to publish the clinical information and the image of the patient in figure 2.

Currently, the age and gender information, as well as the image of the patient in combination with other identifiers, may compromise patient/participant anonymity. If you have not obtained consent to publish, you will have to address this issue by removing the photo of the patient and removing the age or gender information from figure legend 2.

Authors: Written informed consent was obtained from all participants and this has been added in P11L236.

4. We would also like to ask for you to provide more justification for the contributions of FY, as currently they do not automatically qualify for authorship. Contribution to data interpretation, alone, does not usually justify authorship.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

- a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND
- b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND
- c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND
- d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section.

Authors: Authors' names have been represented using their full initials. In addition, F.Y was responsible for data analysis and participated in the interpretation and presentation of the data. This has been provided in P12L245-250.

5. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.
Authors: Funding information has been removed in the acknowledgement. This could be checked in P12L252.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors: We have uploaded a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors.